Clarifictions of Administrative Rules

CSCE 476-876, Spring 2003

URL: www.cse.unl.edu/~choueiry/S03-476-876

Berthe Y. Choueiry (Shu-we-ri) choueiry@cse.unl.edu Tel: (402)476-5444

B.Y. Choueiry 1 Instructor’s notes #2 January 12, 2003

Learning support

• Class time is limited and we have no recitation hour.

Please do your required reading, and do not expect instructor to do (many) exercices in class.

• I will come to class 10-15 minutes ahead of schedule to answer questions.

• We offer plenty of opportunities for individual help.

Please respect office hours.

• Beyond office hours, communicate with us by email as much as possible.

• Do ask for additional help sessions, and

do not miss them when they are offered

B.Y. Choueiry 2 Instructor’s notes #2 January 12, 2003

Feedback mechanisms

• Quizzes are corrected in class.

• Homework and glossaries are promptly corrected. • Grades are listed in great details at board outside my office. • Students who are not performing are contacted directly.

Grades are monitored, but I cannot force you to work. • Your suggestions for improving the course and our feedback mechanisms are welcome, carefully considered, and implemented as quickly as possible.

• Please let us know what other feedback you expect.

B.Y. Choueiry 3 Instructor’s notes #2 January 12, 2003